



# University of Groningen

# Laypeople's Prevalence Estimates of Malingering

Merten, Thomas; Tucha, Lara; Giger, Peter; Niesten, Isabella J.M.; Tucha, Oliver; Fuermaier, Anselm B.M.

Published in: Psychology and Neuroscience

DOI: 10.1037/pne0000303

IMPORTANT NOTE: You are advised to consult the publisher's version (publisher's PDF) if you wish to cite from it. Please check the document version below.

Document Version Publisher's PDF, also known as Version of record

Publication date: 2023

Link to publication in University of Groningen/UMCG research database

*Citation for published version (APA):* Merten, T., Tucha, L., Giger, P., Niesten, I. J. M., Tucha, O., & Fuermaier, A. B. M. (2023). Laypeople's Prevalence Estimates of Malingering: Survey Data from the Netherlands. *Psychology and Neuroscience*, *16*(2), 201-216. https://doi.org/10.1037/pne0000303

Copyright Other than for strictly personal use, it is not permitted to download or to forward/distribute the text or part of it without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license (like Creative Commons).

The publication may also be distributed here under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license. More information can be found on the University of Groningen website: https://www.rug.nl/library/open-access/self-archiving-pure/taverneamendment.

# Take-down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

Downloaded from the University of Groningen/UMCG research database (Pure): http://www.rug.nl/research/portal. For technical reasons the number of authors shown on this cover page is limited to 10 maximum.



© 2023 American Psychological Association ISSN: 1983-3288

2023, Vol. 16, No. 2, 201-216 https://doi.org/10.1037/pne0000303

# Laypeople's Prevalence Estimates of Malingering: Survey Data from the Netherlands

Thomas Merten<sup>1</sup>, Lara Tucha<sup>2</sup>, Peter Giger<sup>3</sup>, Isabella J. M. Niesten<sup>4</sup>, Oliver Tucha<sup>2, 5</sup>, and Anselm B. M. Fuermaier<sup>6</sup>

<sup>1</sup> Department of Neurology, Vivantes Klinikum im Friedrichshain, Berlin, Germany

<sup>2</sup> Department of Psychiatry and Psychotherapy, University of Rostock

<sup>3</sup> Federal Administration, Bern, Switzerland

<sup>4</sup> Clinical Psychology, Open University of the Netherlands

<sup>5</sup> Department of Psychology, National University of Ireland

<sup>6</sup> Department of Clinical and Developmental Neuropsychology, Faculty of Behavioral and Social

Sciences, University of Groningen

**Objective:** Noncredible symptom claims not only occur in a sizable number of forensic cases, but also in clinical contexts. Health providers and forensic experts are confronted with the question of whether individual symptom claims are genuine, grossly exaggerated, or even invented. We investigated laypeoples' prevalence estimates, their attitude and beliefs about malingering. Method: Extending a survey format from a previous Swiss study, we analyzed responses from 975 adult Dutch nationals who responded to an online structured interview on their assumptions and attitudes about malingering (in Dutch language sense of intentional symptom fabrication with an external goal). Results: Participants estimated that a sizeable minority of cases in different prototypical situations feigned symptoms to achieve external goals. They estimated that malingering most commonly occurs in offenders seeking diminished criminal responsibility. One in every seven participants admitted having malingered themselves, and many were confronted with malingering by family members, work colleagues, friends, or neighbors. Although the majority morally condemned malingering, a sizable minority admitted that malingering might be an option for themselves. Participants generally favored an adaptational and a criminological explanatory model of malingering (35% each). Conclusions: The high prevalence of noncredible symptom claims suspected by laypersons in diverse clinical and legal contexts corresponds to base rate estimates derived from numerous other surveys and group studies. Myths perpetuated by some clinical and forensic workers are seriously challenged by laypersons' opinion, based on their very personal social perceptions of the reality, and on their admission of having feigned symptoms in the past themselves.

Lara Tucha D https://orcid.org/0000-0003-3454-2580 Peter Giger D https://orcid.org/0000-0003-0981-6404

Isabella J. M. Niesten D https://orcid.org/0000-0002-4571-8003

This document is copyrighted by the American Psychological Association or one of its allied publishers.

This article was published Online First March 13, 2023. Thomas Merten D https://orcid.org/0000-0001-8423-8507

Oliver Tucha D https://orcid.org/0000-0001-8427-5279 Anselm B. M. Fuermaier D https://orcid.org/0000-0002-2331-0840

This work was supported by the Internet Research Grant

issued by the Faculty of Behavioral and Social Sciences at the University of Groningen. The study was registered with and approved by the ethical review board of the Department of Psychology at the University of Groningen, the Netherlands (Registry Number PSY-2021-S-0033). There are no conflicts of interest to be declared.

Correspondence concerning this article should be addressed to Thomas Merten, Department of Neurology, Vivantes Klinikum im Friedrichshain, Landsberger Allee 49, 10249 Berlin, Germay. Email: thomas.merten@ vivantes.de

#### **Public Significance Statement**

Noncredible symptom presentations (malingered symptomatology) are a major problem both in health care and in forensic referral contexts where significant gain expectations are usually present. Some experts maintain that the problem is negligible, but growing evidence demonstrates the opposite. The results of this survey show that malingered symptom presentations appear to occur on an everyday basis, they are an option for many people, and they should be ruled out proactively by health providers and forensic examiners alike.

*Keywords:* malingering, symptom invention, forensic assessment, attention deficit hyperactivity disorder, survey

Problematic illness presentations, such as factitious disorder, symptom exaggeration, and malingering, pose an enormous burden not only on treatment resources and costs but also on national economies, social security, and welfare systems (e.g., Chafetz & Underhill, 2013; Horner et al., 2014). Among problematic illness presentations, malingering is the one that has attracted most attention in the psychological and medical literature in the last three decades, since the beginning of the era of modern symptom validity assessment. In its early phase, validity assessment was treated as equivalent to malingering assessment, an equation that has been revised on a conceptual level (e.g., Merten & Dandachi-FitzGerald, 2022).

Malingering is the deliberate invention or gross exaggeration of health problems (symptoms), motivated by external incentives (such as monetary compensation, avoidance of punishment, sick leave, early retirement, etc.). In contrast to *pure malingering, partial malingering* refers to an exaggeration of symptoms of a condition that, in fact, exists. In English language conceptualization, *false imputations* are also included in the term malingering. A false imputation is "the attribution of actual symptoms to a cause consciously recognized by the individual as having no relationship to the symptoms" (Resnick et al., 2008, pp. 111–112).

In contrast to other problematic illness presentations (in particular, to factitious disorder), malingering is well-known by laypersons, so much so that it appears to be safe to say that (nearly) *everybody knows what malingering is.* The predecessors of feigned behavioral manifestations are even known in the animal kingdom (such as "injury feigning" by birds, e.g., Cruickshank et al., 1936, or malingering-like behavior in monkeys, e.g., Byrne & Stokes, 2003).

Base rate estimates of malingering continue to be discussed in the literature and they continue to be subject to controversy. Empirical research, survey data included, yield a highly heterogeneous picture, with estimates of malingering, noncredible symptom presentations or invalid test profiles varying from numbers of about 5% in some patient groups (e.g., Martin & Schroeder, 2020; Ruff et al., 2016; Schroeder et al., 2022) to above 50% in some critical forensic referral samples (e.g., Ardolf et al., 2007; Chafetz et al., 2007; Grills & Armistead-Jehle, 2016; Schmand et al., 1998; Streppel & Brusis, 2010). In a review article, Young (2015) judged that "it is difficult to arrive at one percentage or range of percentages that are definite about the proportion of malingering found in forensic disability and related examinations" (p. 196). He proposed a number of about 15% as the most appropriate estimate of malingering, in disability, forensic, and clinical contexts combined. Why it is so difficult, if not impossible, to arrive at reliable prevalence estimates for malingered symptom presentations was discussed in some more detail by Merten and Merckelbach (2020). All we know for sure, it may be said, is that noncredible, invalid symptom presentations occur in a sizable proportion of cases where external gain expectations are an issue.

However, despite all evidence from different regions of the world and a variety of cultural and social-welfare contexts (e.g., Arin, 2012; Quezada-Ortega et al., 2006; Weiss & Rosenfeld, 2010), false myths about malingering prevail in some professional groups and seem to be repeated without accepting evidence from the last three decades of intense research (e.g., González Ordi et al., 2013; Jelicic et al., 2017, for related misconceptions and myths). One of those long-living myths is that "pure malingering" be a rare phenomenon (e.g., Widder et al., 2016), while, in truth, nobody knows how often it occurs. Rather, evidence from large-scale criminal networks with high numbers of invented disability claims reveals that the presentation of deliberately false health problems constitutes a real challenge to social security systems (e.g., Deutsche Rentenversicherung, 2017; Hoffmann, 2019; Kleideiter, 2017, for publicized large-scale organized social security fraud networks). What can be said with some confidence is that gross exaggeration of current or past symptoms is likely to be more frequent than pure malingering of nonexistent illness or injury (e.g., Nies & Sweet, 1994).

In a series of studies, Rogers and his coworkers conceptualized three distinct explanatory models of malingering to which practitioners and researchers may adhere (e.g., Rogers, 1990; Rogers et al., 1998). First, a *pathogenic* model, most prevalent in the early psychodynamic literature, tries to explain malingered symptom presentations as attempts to control actual, underlying psychopathology.

Thus, Klumbies (1980) described a process in which malingered symptom production, which initially constituted a conscious, reflected, goaldirected behavior, may transform into selfdeception. For this process, he coined the term "simulative autosuggestion." At the end point, when autosuggestion was fully functional, Klumbies assumed it to be unconscious and no longer under control by the person's volition. He thought that those cases described by Sigmund Freud as conversion neuroses could be explained by such transformation.

The second model was called *criminological*. It primarily conceived malingering as an expression of antisocial behavior. The criteria of the Diagnostic and Statistical Manual of Mental Disorders, ever since its third edition, DSM-III (American Psychiatric Association, 1980), follow such a conception without notable modification. In contrast to these two models, the seminal work by Rogers (1990) or Rogers et al. (1998) appears to favor an adaptational model to explain (and possibly comprehend) the majority of attempts to feign health complaints. Following this model, malingering is understood as a result of a risk-benefit analysis by the person in question, as their constructive effort to obtain benefits in a given situation while avoiding potential negative consequences.

Because of their high adherence to mostly subjective and unverifiable patient report, clinical

expert opinions in psychotraumatology, pain assessment, and psychosomatic medicine as well as in cases of mild traumatic brain injury and claimed chronic fatigue appear to be particularly vulnerable to mistake invented or grossly exaggerated symptom claims as truthful (e.g., Rickards et al., 2018). Accordingly, Aita et al. (2020) listed chronic pain and posttraumatic stress disorder as "high-risk conditions" for malingering, next to traumatic brain injury, attention deficit hyperactivity disorder (ADHD), and severe psychiatric illness (also cf. Martin & Schroeder, 2020). Clinicians tend to believe the subjective symptom claims and reported history without carefully checking its plausibility, consistency, and possible hidden agendas. Consequently, many diagnoses are primarily, sometimes exclusively, based on selfreported symptomatology. However, such a truth bias is not only costly to health providers and social security systems, but can also cause direct harm to the patient in question (e.g., van der Heide et al., 2020). Against the background of a continued controversy in the German-speaking countries about how to best assess feigned symptom presentations (cf. Schmidt et al., 2011), Schlicht and Merten (2014) performed a pilot study to investigate public opinion about malingering. First, an analysis of 67 newspaper reports available through the Internet revealed a large coverage of cases of malingered health complaints. The articles analyzed for that study frequently resorted to pejorative language and combat rhetoric, apparently aiming to arouse indignation or outrage in the readership. Indeed, as far as German language psychiatric literature is concerned, it appears to continue to significantly adhere to pathogenic models of malingering, but the vocabulary and style used by some authors of published resources also reveals adherence to criminological models (e.g., Hoffmann-Richter et al., 2012).

Second, a small convenient sample of 15 healthy German participants underwent a structured interview with regard to their personal experience with malingering. While the newspaper articles were rated to mostly adhere to the criminological explanatory model, the participants favored an adaptational model of malingering. Laypersons admitted to be largely acquainted with feigned health presentations, either from their own personal experience or from friends, family members, or colleagues. Base rate estimates of malingering in five prototypical contexts (sick leave, early retirement, compensation after a motor vehicle accident, reduced criminal liability, and compensation after criminal assault) yielded mean scores between 46 and 67% of all cases. This means that the participants judged that roughly half to two-thirds of all (German) individuals would try to grossly exaggerate or invent symptoms when being involved in those situations.

Merten and Giger (2018) extended these estimates to a sample of 39 adult Swiss nationals, with very similar results. The prototypical situations were extended to false claims crime-related amnesia and attempts to be exempt from military service (which continued to be compulsory in Switzerland, but not in Germany). Again, participants favored an adaptational model for understanding feigned symptom presentations. Base rate estimates of patients or claimants feigning symptoms in prototypical contexts ranged from 39% (early retirement cases) to 63% (exemption from military service).

Cartwright and Roach (2016) reviewed the responses of 197 laypeople from Britain to the research question of "how likely participants were to engage in different types of malingering" (p. 451), following a hypothetical road traffic accident scenario. Because of a different methodological approach, numbers cannot directly be compared to those of the two studies cited above. Nonetheless, the results showed that a substantial percentage of the participants admitted they might be tempted to engage in either pure or partial malingering or false imputation when claiming depression, posttraumatic stress or whiplash injury as sequelae of an accident. Thus, 20.3% of the participants responded that they were willing to exaggerate insurance claims. In contrast, 9.1% of participants reported that they would be likely to engage in *pure malingering*.

The present study sought to replicate and extend the results of the earlier surveys by Schlicht and Merten (2014) and Merten and Giger (2018) to a larger sample from the Netherlands. Using current online survey methodology, a large sample from the general population could be investigated with a limited employment of resources. The core survey was identical with that used in the two pilot studies, but a supplemental set of items concentrated on feigned ADHD which is one of the main research foci of the research group involved in this study (e.g., Fuermaier et al., 2021; Tucha et al., 2015). With the set of items detailed below, the study aimed to gain more insight into the picture of feigned health complaints from the perspective of nonexperts. This was thought to complement a growing number of studies in which expert views on malingering were surveyed (e.g., Dandachi-FitzGerald et al., 2013; Martin et al., 2015; Mittenberg et al., 2002).

# Method

#### Sample Recruitment and Selection Procedure

Participants were recruited from a pool of panel members registered at PanelInzicht, a Dutch online platform which invites interested members of the public to take part in online studies in exchange for financial reward. Participants received 0.93 Euros (an equivalent of roughly 1 U.S. \$ at that time) for their participation. The questionnaire was designed and implemented with Qualtrics survey software and invited adults aged 18-65 years from the Dutch community between 18 June and 5 July, 2021. An initial sample of 1,339 participants accepted the invitation and started working on the survey. Eighty participants from this sample were excluded from further analysis because they completed less than two-thirds of the survey questions. This exclusion resulted in 1,259 participants with a majority of responses given which were considered in further analyses.

Because online survey responses are known to be particularly vulnerable to careless responding (e.g., Ward & Meade, 2022), three experimental compliance checks were implemented into the survey at different points. Participants were given three basic mathematical items (e.g., assume that 20% of a total group of 200 persons feign symptoms. How many persons would feign symptoms?). This type of percentage problems was chosen because they had to yield part of their estimates in percent, so a basic understanding of the mathematics was a prerequisite for valid responding. It turned out that a substantial number of participants gave nonsense or highly suspect or inconsistent answers (e.g., "30,000,000," "I am unable to calculate anything," or "ijhgfrtyjk"). Thus, as an inclusion criterion, participants were required to solve at least two of the three tasks. On this basis, 213 of the 1,259 participants were excluded, resulting in a sample size of 1,046. Finally, the individual times to complete the survey was checked. It varied between less than 3 min to more than 6 days, with a highly skewed distribution and a (plausible) medium score of 11.6 min. On this basis, all response times below 5 min and above 120 min were judge to be suspect of an unreliable response behavior. This led to the exclusion of the fastest 3.8% and the slowest 3.0% of the sample.

After the selection procedure, we had full response patterns for the first part (Table 1) and incomplete responses in the second part (ADHD questions, Table 2) for nine participants.

# Table 1

No.

<i>Questions Presented to the Participants in Online F</i>	'ormat
--	--------

- 1. What percentage of all Dutch people, do you think, malinger symptoms (make up symptoms<sup>a</sup>)?
- Have you ever been in contact with people who malingered symptoms, such as family members, friends, colleagues, neighbors, etc.?

Contents

- Have you ever malingered any symptoms yourself?—Specify: physical, mental, or both kinds of symptoms. —Specify: which symptoms?
- 4. What percentage of all Dutch people, do you think, would malinger symptoms to obtain an advantage in the following situations:
  - 1. Asking for a sick leave certificate from the general practitioner?
  - 2. In an independent medical examination, seeking early pension.
  - 3. Having been victim of a motor vehicle accident and seeking compensation for presumably associated health problems.
  - 4. Having been victim of a crime and seeking compensation for presumably associated health problems.
  - Having committed a criminal offence and seeking diminished legal responsibility or exemption from punishment.
     Seeking prescription of a medical drug.
- 5. Do you belief that you may feign symptoms in the following situations? (a to f from Question 4)
- 6. [Only for participants who reported having feigned health problems.]
  - How do you feel when you feign symptoms?

[Only for participants who reported not having feigned symptoms in the past.]

How would you feel in a situation where you feigned symptoms?

- 1. I feel or would feel guilty.
- 2. I feel or would feel thrilled.
- 3. I feel or would feel satisfied with the result.
- 4. I feel or would feel neutral.
  - (multiple responses possible)
- 7. In the following situations, how would you judge the act of malingering symptoms to achieve a certain goal, in moral-ethical terms?
  - 1. In general
  - 2. To get a sick leave from the general practitioner.
  - 3. In an independent medical examination, seeking early pension.
  - 4. Being victim of a motor vehicle accident and seeking compensation for presumably associated health problems.
  - 5. Being victim of a crime and seeking compensation for presumably associated health problems.
  - 6. Seeking diminished legal responsibility or exemption from punishment, after having committed an offence.
  - 7. Seeking prescription of a medical drug.
  - Chose one responses for every situation: very bad-bad-no opinion-good-very good
- 8. Now, imagine persons who are malingering symptoms. What percentage of attempts to malinger symptoms can be explained by the following constellations?
  - (The three given number must sum up to 100.)
    - It is an act of fraud committed in a cool and unscrupulous manner. %
       The behavior of the person in question reveals a mental disorder (because a healthy person would not do % something like this).
    - 3. The person in question is in a situation in which this appears to be the best behavioral option. %
- 9. In some situations (such as early pension claims, compensation seeking, claimed diminished criminal responsibility), experts are employed to examine the physical and mental health of the claimants.
  Numinaging that a patient who is mpling up supertanging during the physical physical and mental health of the claimants.
  - Now imagine that a patient who is making up symptoms is examined by an expert. How often, do you think, % experts will be able to detect the malingering?
- 10. Now imagine that a patient is suffering from genuine symptoms. How often, do you think, experts will % misdiagnose the real symptoms as feigned symptoms?

<sup>&</sup>lt;sup>a</sup> See Footnote 1 in the text for understanding that only one aspect of *malingering*, symptom invention or fabrication, is grasped by the Dutch *simulatie*.

# **Participants**

After application of these steps of plausibility and reliability checks of the data set, the resulting final sample comprised N = 975 respondents. These were 370 men (38%) and 605 women with a mean age of 50.6 years (SD = 11.1) and an age span between 18 and 65 years.

Education was assessed with self-reported highest level of achievement based on the Dutch Education System. Eleven participants (1.1% of N = 975) reported having completed primary education as highest educational degree, 104 (10.7%) participants indicated prevocational education, while 329 (33.7%) participants additionally completed vocational training. Ninety-four (9.6%) participants successfully completed higher secondary general education, and 42 (4.3%) participants obtained higher secondary prescientific education. Two hundred fifty-eight participants (26.5%) indicated to have a bachelor degree, and 126 (12.9%) reported to have a master degree (both including Universities and Universities of Applied Science). Eleven (1.1%) participants reported to have a PhD degree.

## Material

First, the participants were asked to give their personal definition of malingering.<sup>1</sup> This was followed by the explanation that, in medicine, malingering of symptoms (in Dutch: *het simuleren van symptomen*) was defined as the conscious invention or fabrication of symptoms with the intention to gain a defined goal, such as sick leave, exemption from unpleasant work or tasks, exemption from military service, early pension, monetary compensation, or the prescription of a medical drug. Then, the questions were given, as presented in Table 1.

This catalog of questions was adapted from the Merten and Giger (2018) study in Switzerland. As in that study, adherence to the explanatory models of malingering (Rogers, 1990) was assessed by presenting core statements of the three models (e.g., "The behavior of the person in question reveals a mental disorder, because a healthy person would not do something like this"; for a quintessential description of the pathogenic model, see Table 1).

This was followed by a number of additional questions related to malingered adult ADHD (its frequency, advantages and disadvantages; Table 2). The survey finished with the request to the participants to imagine persons who were malingering adult ADHD and compare them with nonmalingerers. The question was if the ADHD malingerers were thought to be involved more frequently in other acts of malingering (using the same prototypical situations as in the first part of the interview). A 5-point Likert format was used.

# Results

The percentage of Dutch people who have been engaged in malingering (in the above Dutch language sense of symptom invention) was judged, on average, to be as high as 31.2%, that is one in every three persons (SD = 19.8; Mdn = 29%; Question 1). The range of given answers varied between the extremes of 0% (nobody) to 100% (everybody). About 22.7% of the participants (of N = 975, for all numbers given for the first part of the interview, Table 1) reported having had contact with people malingering symptoms within their families, 24.9% among their friends, 38.9% among their colleagues, and 15.4% in their neighborhoods (Question 2).

About 14.3% of participants admitted that they had feigned symptoms themselves at some point in the past (Question 3). Of those respondents who specified the chosen symptomatology, the majority (55%) had malingered physical symptoms, a minority (7.4%) mental symptoms, and a sizable proportion (37.5%) both physical and mental symptoms.

With regard to prototypical constellations (Question 4), the responses are summarized in Table 3. Almost half of the respondents would expect most invented symptomatology in the context of avoiding criminal responsibility. Although a nontrivial, sizable minority of cases was judged to be linked with malingered symptoms in all prototypical constellations, the estimates were lower than those obtained in the previous studies in Germany and Switzerland. The same is true for the question, if faking-bad would be an option for the participants themselves if they were in such a situation (Question 5). The highest rate of affirmation was obtained for compensation seeking in personal injury cases,

<sup>&</sup>lt;sup>1</sup> For understanding the international literature, it is important to know that, similar to some other languages, the Dutch term *simulatie* as well as the German term *Simulation* are equivalent to that aspect of malingering that covers the invention or fabrication of symptoms, not just a gross symptom exaggeration, and not false imputations. In this sense, when asked about personal experience and estimated prevalence rates of *simulatie*, the responses mostly cover the domain of symptom invention.

Questions About Attention Deficit Hyperactivity Disorder Presented to the Participants in Online Format

No.	Contents
11.	Have you ever heard of attention deficit hyperactivity disorder (ADHD) in adults?
12.	Have you ever considered malingering ADHD yourself?
13.	<ul> <li>Do you know anyone who has received an ADHD diagnosis, but is actually pretending to have or grossly exaggerating his/her symptoms?</li> <li>Yes, this is a student.</li> <li>Yes, this is not a student.</li> <li>Yes, I know of both.</li> <li>No.</li> </ul>
14.	What percentage of students with an ADHD diagnosis, do you think, does in actuality not have ADHD because he/she malingers their symptoms?
15.	What percentage of all adults (excluding students) with an ADHD diagnosis, do you think, does in actuality not have ADHD because he/she malingers their symptoms?
16.	Do you think it is easy to malinger ADHD symptoms?
17.	Do you think there are advantages to malingering ADHD symptoms? Yes (namely:)/No
18.	Do you think there are <i>disa</i> dvantages to malingering ADHD symptoms? Yes (namely:)/No
19.	Have you ever used stimulant medication (i.e., medication for ADHD) without a prescription from a medical doctor?
20.	Do you know a student who is using (or has used) stimulant medication (i.e., medication for ADHD) without a prescription from a medical doctor?
21.	Do you know anyone who is not a student but is using (or has used) stimulant medication (i.e., medication for ADHD) without a prescription from a medical doctor?
22.	What percentage of experts can, in your opinion, accurately judge whether ADHD symptoms are genuine or malingered?
23.	<ul><li>Imagine that you would compare people who malinger ADHD symptoms with people who do not malinger ADHD symptoms. Do you think that people who malinger ADHD symptoms would more often or less often also malinger symptoms in the following situations?</li><li>1. In general.</li></ul>
	2. To get a sick leave declaration from the general practitioner.
	3. In an independent medical examination, seeking early pension.
	4. When being victim of a motor vehicle accident and seeking compensation for presumably associated health problems.
	<ol> <li>5. When being victim of a crime and seeking compensation for presumably associated health problems.</li> <li>6. Seeking diminished legal responsibility or exemption from punishment, after having committed an offence.</li> <li>7. Seeking prescription of a medical drug.</li> </ol>
	Chose one responses for every situation: far less often—less often—equally often—more often—far more often

caused either by an offender (44.6%) or by a motor vehicle accident (39.9%).

Participants who had reported having feigned health problems in the past mostly reported feelings of guilt (80.6%), but 16.5% of them also admitted feelings of thrill (Question 6). Satisfaction with the result of malingering was reported by 46.0% of the respondents while 39.1% reported having felt neutral. Those participants who had not responded having malingered in the past themselves thought malingering would mostly be accompanied by feelings of guilt (91.6%). About one in every five respondents reported expectations of feeling neutral (21.7%) or satisfied with the results (22.9%), while only 7.1% expected to feel thrilled in the context of feigning health problems. For all four emotional categories, the differences between the two subgroups were significant (separate crosstabulation testing, with  $\chi^2$ from 21.90 to 42.28, all p < .001).

With regard to the moral dimension of malingering (Question 7), 80.5% of the respondents chose a negative category (bad or very bad), while only 1.6% of the respondent judged such behavior to be good or very good. There was some variation among the specified prototypical situations, with the least severe moral judgement for health problems malingered by victims of a criminal offence (57.6% bad or very bad, 9.4% good or very good) to the lowest scores of acceptance for offenders who seek diminished criminal responsibility (82.7% bad or very bad, 2.3% good or very good).

Table 4 presents the results of the participants' responses about how to assess malingering

# Table 3

Prevalence of Malingering Estimates Given by the Participants for Different Referral Contexts (Questions 4 and 5 of the Interview)

	Estima	ted prevalence	e of malingering (%)		
Referral domain/referral question	Range	M (SD) [Mdn]	Comparison values from two previous studies <sup>a</sup>	Malingering might be an option for the respondent (%)	Percentages reported in two previous studies (%) <sup>a, b</sup>
Clinical: obtaining sick leave from the general practitioner	0–100	30.0 (21.8)	55.9 (22.2) 41.9 (29.2)	26.6	66.6 20 5
Clinical: prescription of a drug	0–100	32.6 (23.9) [27.0]		25.9	
Civil forensic: early pension	0–100	23.6 (21.5)	46.2 (26.3) 39.0 (27.7)	18.7	60.0 23.1
Civil forensic: compensation after motor vehicle accident	0–100	36.4 (24.3) [30.0]	49.1 (28.1)	39.9	53.3 46.2
Criminal forensic: diminished criminal responsibility	0–100	47.4 (28.4) [47.0]	66.8 (22.1) 60.6 (26.8)	35.9	66.6 56.4
Criminal forensic: crime-related	—	_	60.9 (28.3)	—	51.3
Forensic—civil and social legislation: compensation for victims of a criminal act	0–100	36.1 (25.2) [30.0]	53.2 (22.7) 49.2 (27.1)	44.6	66.6 59.0
Military forensic: exemption from military service	—	—	63.0 (28.0)	—	41.0

<sup>a</sup> Comparison data from a German pilot study (Schlicht & Merten, 2014; upper line) and data from Switzerland (Merten & Giger, 2018; lower line). <sup>b</sup> Column refers to the question if malingering might be an option for the participants themselves.

behavior in terms of Rogers' explanatory models of malingering (Question 8). While the majority of malingered presentations was judged to be explainable using an adaptational model (35.3%), they thought a criminological explanatory model would apply to nearly as many cases (34.8%), with somewhat less preference for the pathogenic model (29.8%).

The ability of experts to correctly identify fabricated or grossly exaggerated symptom presentations was judged to be poor (Questions 9 and 10). The participants thought on average that only 58.4% of malingered symptom presentations (SD = 23.3; Mdn = 61.0%) were correctly identified by doctors, that is 41.6% were not detected. In contrast, a high number (34.3%, SD = 22.5; Mdn = 30.0) of true symptomatology was thought to be mistaken by experts as malingered presentations (Table 5).

With regard to adult ADHD, 87.3% of the participants responded they had heard of this disorder before (Question 11). Nine participants (0.9% of N = 975) had themselves considered to feign adult ADHD (Question 12). Eighty-seven participants (8.9% of N = 975) responded they knew people who had the diagnosis of adult ADHD

## Table 4

Layperson's Attribution of Malingered Symptomatology to Three Explanatory Models (Rogers, 1990), Question 8 of the Interview

				Comparative data		
Model	Range	M (%)	SD	Merten and Giger (2018)	Schlicht and Merten (2014)	
Criminological Pathogenic Adaptational	0–100 0–100 0–100	34.8 29.8 35.3	31.4 18.6 21.7	32.7% (18.1) 25.7% (16.2) 41.6% (20.9)	27.1% (20.5) 20.3% (10.9) 52.5% (21.3)	

Table	5
-------	---

Estimated Error Rate of	f Experts in Thei	r Detection of	f Malingered	Symptoms	(Questions 9	∂ and 10 of
the Interview)						

	Current results			Comparative data		
False classifications	Range	M [Mdn]	SD	Merten and Giger (2018)	Schlicht and Merten (2014)	
Wrongly confirmed malingered symptoms (false-positive rate)	0-100	34.3% [30.0%]	22.5	23.4% (17.3)	34.3% (24.1)	
Wrongly missed malingering (false-negative rate)	0–100	41.6% [39.0%]	23.3	33.4% (21.9)	35.9% (18.5)	

but behaved in a way as if they would seriously exaggerate their symptoms (Question 13).

As far as university students were concerned (Question 14), participants thought that 22.4% (of N = 975) of those with the diagnosis of ADHD had fabricated their symptoms (SD = 18.3; Mdn =19.0), in contrast to 19.5% (SD = 16.8; Mdn =14.0) in the rest of the population (nonstudents; Question 15). One-third of the respondents (33.3%, n = 324 of 974) thought it was easy to malinger adult ADHD (Question 16). Only onefifth (21.7%, n = 211 of 974) of the respondents expected that feigning adult ADHD would be connected with advantages (such as better concentration in healthy people, too; extra time on exams; access to psychostimulants, free of cost; Question 17) while 69.0% (n = 672 of 974) expected that such a behavior would rather bring about disadvantages (such as negative side effects of medication; negative impact of wrong diagnosis; moral dilemma, feelings of guilt, etc.; Question 18).

About 2.1% of the participants (n = 20 of 973 valid responses) admitted having consumed psychostimulants without a medical prescription (Question 19) while 6.2% (n = 60) reported knowing a student who was using psychostimulants without a prescription (Question 20). 3.8% (n = 37) reported knowing someone who was not a student and consumed psychostimulants (Question 21). The number of experts who are able to accurately judge whether ADHD symptoms are genuine or malingered (Question 22) was expected to be poor, with an estimate of only 56.4% (SD = 25.2; Mdn = 60.0).

Persons who malingered adult ADHD were expected to be involved more frequently in other acts of malingering (Question 22; Table 6). Primary at-risk contexts were, according to the participants' judgment, seeking diminished criminal responsibility after having committed an offence and, directly related to malingered ADHD, seeking unjustified prescription of a medical drug.

# Discussion

Ever since the seminal survey by Mittenberg et al. (2002), the international literature has brought about a notable number of expert surveys investigating the prevalence of malingered, feigned, or otherwise invalid symptom presentations both among forensic and clinical patient groups (e.g., Dandachi-FitzGerald et al., 2013; Martin et al., 2015; Santamaría et al., 2013). More recently, a number of surveys from different countries (e.g., from the U.S.: Aita et al., 2020; from the Netherlands: Dandachi-FitzGerald et al., 2020; from Italy: Giromini et al., 2022; from Spain: Puente-López et al., 2022) confirm our main result that noncredible symptom presentations are perceived by experts to occur in a sizable, nontrivial number of cases. Also, in more recent years, noncredible symptom reports in clinical and rehabilitation contexts have increasingly drawn the attention of researchers and practitioners alike (e.g., Carone & Bush, 2018; Martin & Schroeder, 2020; Merten et al., 2020; Schroeder & Martin, 2022). In contrast, some medical and psychological experts appear to continuously adhere to a position accepting clinical patient presentations as genuine, without questioning their validity at all. In this vein, Reuber et al. (2005) had remarked that: "Fortunately, it is rarely necessary for a clinician to determine whether symptoms are intentional" (p. 308, emphasis *added*). Such a position, although avoiding effort and possible conflict for the clinician, may also be highly detrimental to the patient. This is often the case in patients with factitious disorder, to name the most obvious constellation of potential harm unnecessarily caused to mental health patients.

	Expected frequer other refer	5-point Likert scale <sup>a</sup>		
Context	Much less or less frequently (%)	Much more or more frequently (%)	М	SD
In general	17.8	36.1	3.19	0.83
Sick leave from general practitioner	10.0	53.1	3.50	0.80
Early pension	16.4	39.2	3.24	0.86
Compensation claim after MVA	12.9	47.6	3.41	0.86
Compensation claim as crime victim	11.1	49.5	3.46	0.84
Diminished criminal responsibility	9.2	56.4	3.61	0.89
Prescription of a drug	6.0	59.4	3.72	0.87

Table	6
-------	---

Estimated Probability to Malinger in Other Contexts

*Note.* MVA = motor vehicle accident. Comparison of adults who feign attention deficit hyperactivity disorder with adults who do not.

<sup>a</sup> 5-point Likert scale, from 1 (*much less frequently*) to 5 (*much more frequently*), with the point of indifference at 3 (*equal frequency*).

Thus, van der Heide et al. (2020) published two case vignettes in which nondetected feigned symptomatology resulted in potentially harmful interventions. The detrimental effects of false diagnoses in cases of claimed ADHD and other incentivized nonvisible disabilities, their medicalization and overpathologizing have more recently been discussed by Suhr and Johnson (2022) and Harrison (2022).

In the present study, we surveyed Dutch laypeople's beliefs and convictions about malingered health problems using a final sample of 975 participants responding to a set of questions via an Internet research platform. As in two previous studies from Germany and Switzerland (Merten & Giger, 2018; Schlicht & Merten, 2014) with low numbers of participants, the first important result was that a substantial number of respondents admitted having feigned symptoms themselves (14.3%) in the past or knowing people within their families (22.7%), among their friends (24.9%), their colleagues (38.9%), or neighbors (15.4%) who had been engaged in malingering. This result contradicts myths continuously held among some clinicians and even forensic experts that malingering was a rare or even negligeable event (e.g., Ramesh, 2013). However, the number of participants who admitted having feigned symptoms themselves at some point in the past (14.3%)turned out to be relatively low, in contrast to results obtained in other studies (Dandachi-FitzGerald et al., 2020: 34.2%; Merten & Giger, 2018: 41.0%; Puente-López et al., 2022: 33.9%). The most plausible explanation for this discrepancy is

the definition of the Dutch *simulatie* to be limited to pure malingering only.

The participants estimated the base rate of malingered symptom presentations in the range of 31.2%. This number is well in line with results from various expert surveys, in particular those estimates published by Mittenberg et al. (2002) for personal injury cases (adjusted base rate: 34.4%) and disability or worker's compensation referrals (32.7%). Other expert surveys yielded similar estimates of noncredible symptom presentations or malingering (e.g., Aita et al., 2020: up to 29.3% of chronic pain patients; Dandachi-FitzGerald et al., 2013: 20% of forensic neuropsychological patients; Puente-López et al., 2022: 20 to 40% of forensic psychological cases). They also compare with the results of another study from the Netherlands. In the Dandachi-FitzGerald et al. (2020) study, 34% of their participants "... said that they were certain that a person they knew had feigned symptoms because the person in question had told them so" (p. 227).

Our participants assumed the highest prevalence to occur for the referral question of diminished criminal responsibility (47.4%), the lowest in persons with early pension claims (23.6%). A sizable minority of participants also reported that malingering would be an option for themselves in one of the described prototypical contexts. Being an innocent victim of circumstances or adverse events appears to facilitate the readiness of people to engage in manipulated symptom presentations, confirmed by 44.6% of the participants if they were victim of a criminal offense and 39.9% of them in the context of motor vehicle accidents not caused by themselves. It seems as if being a victim is considered as a moral justification to compensation. In this context, parallels to reactions of embitterment as described by Linden and Rotter (2020) are obvious. Thus, Linden et al. (2020) described feelings of injustice and humiliation as the most frequent and most hurting stressors in psychosomatic patients. At the same time, it may be assumed that those feelings can smooth the contrast between objectively unjustified compensation claims and deeply rooted subjective convictions that such compensation awards would be fully justified. Iverson et al. (2018) found that perceived injustice in patients with mild traumatic brain injury, among them victims of motor vehicle accidents and assaults, was not only associated with more reported postconcussive, depressive, and pain symptoms, but also with a higher probability of failing on performance validity tests. In an Australian sample of emergency department trauma cases, return to works after compensable injury was significantly mediated by perceived injustice (Giummarra et al., 2017). At a different level, Cartwright and Roach (2016) found in the context of motor vehicle accidents that exaggerated (partially malingered) compensation claims are considered to be morally "less of a crime as it is something everybody does" (p. 417).

This corresponds to a further result of our study. A majority of the respondents (80.5%) morally characterized malingering in negative terms (as bad or very bad). However, when they were asked to specify their judgments according to prototypical situations, only 57.6% did so with respect to victims of a criminal offence, 63.7% with respect to victims of a motor vehicle accident, but 82.7% when diminished criminal responsibility was sought.

At a level of emotional reactions, a high percentage of participants reported predominant feelings of guilt in a situation where they had malingering health complaints or imagined they would do so. However, those who had, in fact, malingered symptoms in the past confirmed feelings of guilt more often than those who had not (91.6% vs. 80.6%). Nevertheless, almost half of those who had been engaged in malingering in the past, reported they had felt satisfied with the results. In a broader context, it might be interesting to know if personality traits or social factors are associated with feigned symptom presentations, underperformance in psychological assessment, or overreporting with respect to symptom claims (cf. van Helvoort et al., 2022).

With regard to explanatory models of malingering as described by Rogers (1990), most participants adhered to the adaptational and the criminological models while both previous studies had shown a clear preference for the adaptational model. In all three studies, the pathogenic model ranged lower in participants' preference. The dominance of the adaptational model corresponds to a shift in experts' conceptualization of malingering from the 1970s and 1980s (when malingering was understood as being regularly coupled with antisocial personality) to the present times. Adherence to an adaptational model is also able to ensure the maintenance of a positive selfconcept. Thus, a victim of a criminal offense or a motor vehicle accident may feel morally entitled to an appropriate compensation, even if this means that the effects of the offense or the accident will be exaggerated or (partly) fabricated. In this context, an association between malingering and cognitive dissonance may be interesting (Merckelbach & Merten, 2012).

As in the two previous studies, the ability of experts to reliably distinguish between genuine and feigned symptom presentations was judged to be poor, both in terms of false-positive and false-negative determinations. The particularly high rate of 41.6% of failed malingering detection estimated in our study may (at least partially) be based on the personal experience of the participants or observations in their social environment, friends, neighbors, colleagues, or family members, of how easy it appears to convince a doctor to accept an invented or grossly exaggerated symptom report as genuine. The high rate of false-positive determinations of 34.3% appears to be as problematic, if not more so. In symptom and performance validity assessment, a reduction of false-positive errors is considered to be a prime concern, due to the potential negative effects of incorrectly labeling a genuine symptomology as noncredible or feigned. In laypeople's perspective, the ability of experts to reliably distinguish between genuine and feigned ADHD appears to be similarly poor. This is well in line with the description of multiple difficulties in determining genuine ADHD, the large variety of diagnostic approaches used by practitioners in real-world assessment contexts (e.g., Schneider et al., 2019), and calls for a more systematic use of validity

tests in clinical ADHD assessments (Suhr & Johnson, 2022).

In fact, results from early malingering research demonstrated how difficult it is for experts and would-be experts to reliably distinguish between true and false contents of communication in general, and genuine and fabricated symptom reports or test profiles in particular (e.g., Ekman & O'Sullivan, 1991; Faust et al., 1988; Heaton et al., 1978; Rosenhan, 1973). The use of modern symptom and performance validity testing and a comprehensive analysis of consistency and plausibility will have improved the situation considerably, but such an improvement can only occur when these means of deception detection are taken seriously.

We also included some additional questions to the survey specifically focusing on malingered ADHD. Fabricated claims of ADHD are a problem of special attention in particular in younger adults such as university students, with external gain expectations directed either at the prescription of psychostimulants or at academic accommodations. Consequently, this problem has drawn much attention in the research literature of the last decade (e.g., Bryant et al., 2018; Fuermaier et al., 2021; Hurtubise et al., 2017; Marshall et al., 2016; Martin & Schroeder, 2020; Tucha et al., 2015). The participants of our survey appeared to be aware of this problem. They estimated that almost a quarter of all university students with claimed ADHD were malingering.

It appears to be noteworthy that the participants of the survey judged that individuals who feigned ADHD were more likely to get involved in acts of malingering in different contexts. This item (Question 23) was included in the study to gain some preliminary insight into laypeople's beliefs about links between person-related factors and dishonest behavior. Historically, antisocial personality had been conceived to be closely linked to malingering. This is documented by the continuous adherence of the Diagnostic and Statistical Manual (DSM-5; American Psychiatric Association, 2013) to its malingering criteria which have not been given up for more than three decades. More recent research found generally weak correlations between personality traits and overreporting of symptoms (cf. van Helvoort et al., 2022, for a recent review). On a more general level, Palena et al. (2022) described the link between personality traits, moral disengagement, Machiavellianism, and lying tendencies. The question if persons who are engaged in one act of feigned health presentations are indeed more likely to engage in more acts, will be an interesting topic to be researched in the future.

Compared to a recent survey among university students in the Netherlands (Fuermaier et al., 2021), the present survey among the general Dutch population showed a lower degree of suspicion toward the veracity of the diagnosis of adult ADHD. This was also reflected in a lower number of participants who admitted knowing someone who was feigning ADHD.

The first and possibly major limitation of the study is that it was performed as an Internet online study with a substantial percentage of participants with incomplete and assumably invalid response patterns. Participation was paid for and we have no information what factors determined registration at the platform and motivation to do this particular survey. The main advantages of using online survey formats are their cost effectiveness, convenience, and the possibility of studying a large number of participants in a limited time frame, but representativeness of the sample remains a major source of concern (e.g., Andrade, 2020; Klieve et al., 2010). A possible selection bias for participation has to be kept in mind when evaluating our results. At the same time, it has to be kept in mind that other survey methods like convenience sampling or recruitment of participants through the social media face similar restraints.

Another limitation is the conceptual difference between the English language concept of malingering which comprises symptom fabrication (pure malingering), gross exaggeration of symptoms (sometimes referred to as partial malingering) and false imputations. As is the case for some other languages like German or Spanish, the Dutch noun simulatie only refers to the first aspect (i.e., symptom invention or fabrication), at least in everyday language and from laypeople's perspective. In contrast, medical and psychological research often refers to the English language concept of *malingering* comprising both symptom fabrication and gross exaggerations of symptoms (e.g., Schmand & Ponds, 1997). The questions in the studies of Schlicht and Merten (2014) in Germany and Merten and Giger (2018) in Switzerland had explicitly included such gross exaggerations in the questions posed to the participants. Although the participants of the present study had received an introduction alerting them to the inclusion of exaggeration of symptoms, it will remain unclear how far they had kept this in mind when responding to the items of the survey. This factor may explain some of the differences in the results of the three studies, but another important factor is the fact that both the German and the Swiss studies were pilot studies with low numbers of participants.

Despite these limitations, the main result of all three studies appears to confirm that malingering of health problems occurs at a nontrivial rate in different contexts, including clinical presentations, a result that corresponds to the findings of most expert surveys and those empirical studies that include modern methods of validity assessment. Modern validity research came to the conclusion, as formulated by Sweet (2009), that "... failure to proactively assess for possible malingering in a forensic case is now considered below the standard of acceptable practise" (p. 6). It is a more recent development that the same demand should be extended to clinical and rehabilitation patients whenever substantial external gain expectations can be identified or medical complaints remain unexplained (e.g., Martin & Schroeder, 2020).

# References

- Aita, S. L., Borgogna, N. C., Aita, L. J., Ogden, M. L., & Hill, B. D. (2020). Comparison of clinical psychologist and physician beliefs and practices concerning malingering: Results from a mixed methods study. *Psychological Injury and Law*, *13*(3), 246– 260. https://doi.org/10.1007/s12207-020-09374-x
- American Psychiatric Association. (1980). *Diagnostic* and statistical manual of mental disorders (3rd ed.).
- American Psychiatric Association. (2013). Diagnostic and statistical manual of mental disorders (5th ed.). Andrade, C. (2020). The limitations of online surveys.
- *Indian Journal of Psychological Medicine*, 42(6), 575–576. https://doi.org/10.1177/0253717620957496
- Ardolf, B. R., Denney, R. L., & Houston, C. M. (2007). Base rates of negative response bias and malingered neurocognitive dysfunction among criminal defendants referred for neuropsychological evaluation. *The Clinical Neuropsychologist*, 21(6), 899–916. https://doi.org/10.1080/13825580600966391
- Arin, N. (2012). Performance on the Test of Memory Malingering among Thai forensic and non-forensic samples [Article in Thai language]. *Journal of Clinical Psychology*, 44(2), 47–60.
- Bryant, A. M., Lee, E., Howell, A., Morgan, B., Cook, C. M., Patel, K., Menatti, A., Clark, R., Buelow, M. T., & Suhr, J. A. (2018). The vulnerability of selfreported disability measures to malingering: A simulated ADHD study. *The Clinical Neuropsychologist*, 32(1), 109–118. https://doi.org/10.1080/13854046 .2017.1346145

- Byrne, R. W., & Stokes, E. (2003). Can monkeys malinger? In P. W. Halligan, C. Bass, & D. A. Oakley (Eds.), *Malingering and illness deception* (pp. 54–67). Oxford University Press.
- Carone, D. A., & Bush, S. S. (2018). Validity assessment in rehabilitation psychology and settings. Oxford University Press.
- Cartwright, A., & Roach, J. (2016). Fraudulently claiming following a road traffic accident: A pilot study of U.K. residents' attitudes. *Psychiatry, Psychology, and Law*, 23(3), 446–461. https://doi.org/10.1080/ 13218719.2015.1080148
- Chafetz, M. D., Abrahams, J. P., & Kohlmaier, J. (2007). Malingering on the Social Security disability consultative exam: A new rating scale. Archives of Clinical Neuropsychology, 22(1), 1–14. https:// doi.org/10.1016/j.acn.2006.10.003
- Chafetz, M., & Underhill, J. (2013). Estimated costs of malingered disability. Archives of Clinical Neuropsychology, 28(7), 633–639. https://doi.org/10.1093/ arclin/act038
- Cruickshank, A. D., Rutter, R. J., & Cottam, C. (1936). "Injury feigning" by birds. *The Auk*, *53*(4), 478–482. https://doi.org/10.2307/4078314
- Dandachi-FitzGerald, B., Merckelbach, H., Bošković, I., & Jelicic, M. (2020). Do you know people who feign? Proxy respondents about feigned symptoms. *Psychological Injury and Law*, 13(3), 225–234. https://doi.org/10.1007/s12207-020-09387-6
- Dandachi-FitzGerald, B., Ponds, R. W. H. M., & Merten, T. (2013). Symptom validity and neuropsychological assessment: A survey of practices and beliefs of neuropsychologists in six European countries. Archives of Clinical Neuropsychology, 28(8), 771–783. https://doi.org/10.1093/arclin/act073
- Deutsche Rentenversicherung. (2017). Deutsche Rentenversicherung begrüßt erste Urteile gegen "Rentenbetrüger" [The German Pension Fund welcomes first verdicts against "pension fraudsters"]. Retrieved November 01, 2017, from http://www.deutsche-re ntenversicherung.de/Westfalen/de/Inhalt/4\_Presse/ Journalisten/Pressemitteilungen/2017/Urteil\_gegen\_ Rentenbetrueger\_11\_10\_2017.html
- Ekman, P., & O'Sullivan, M. (1991). Who can catch a liar? American Psychologist, 46(9), 913–920. https:// doi.org/10.1037/0003-066X.46.9.913
- Faust, D., Hart, K. J., Guilmette, T. J., & Arkes, H. R. (1988). Neuropsychologists' capacity to detect adolescent malingerers. *Professional Psychology, Research* and Practice, 19(5), 508–515. https://doi.org/10.1037/ 0735-7028.19.5.508
- Fuermaier, A. B. M., Tucha, O., Koerts, J., Tucha, L., Thome, J., & Faltraco, F. (2021). Feigning ADHD and stimulant misuse among Dutch university students. *Journal of Neural Transmission*, 128(7), 1079– 1084. https://doi.org/10.1007/s00702-020-02296-7
- Giromini, L., Pasqualini, S., Coriat Loia, A., Pignolo, C., Di Girolamo, M., & Zennaro, A. (2022). A survey

of practices and beliefs of Italian psychologists regarding malingering and symptom validity assessment. *Psychological Injury and Law*, *15*(2), 128–140. https://doi.org/10.1007/s12207-022-09452-2

- Giummarra, M. J., Cameron, P. A., Ponsford, J., Ioannou, L., Gibson, S. J., Jennings, P. A., & Georgiou-Karistianis, N. (2017). Return to work after traumatic injury: Increased work-related disability in injured persons receiving financial compensation is mediated by perceived injustice. *Journal of Occupational Rehabilitation*, 27(2), 173–185. https://doi.org/10 .1007/s10926-016-9642-5
- González Ordi, H., Santamaría Fernandez, P., & Capilla Ramírez, P. (2013). Mitos y concepciones erróneas en torno a la evaluación de la simulación y la validez de los síntomas [Myths and false conceptions about malingering assessment and symptom validity]. *Infocop*, 62(1), 7–11.
- Grills, C. E., & Armistead-Jehle, P. (2016). Performance validity test and neuropsychological assessment battery screening module performances in an active-duty sample with a history of concussion. *Applied Neuropsychology. Adult*, 23(4), 295–301. https://doi.org/10 .1080/23279095.2015.1079713
- Harrison, A. G. (2022). Attention deficit hyperactivity disorder, learning disorder and other incentivized diagnoses: A special issue for psychologists. *Psychological Injury and Law*, 15(3), 227–235. https:// doi.org/10.1007/s12207-022-09460-2
- Heaton, R. K., Smith, H. H., Jr., Lehman, R. A. W., & Vogt, A. T. (1978). Prospects for faking believable deficits on neuropsychological testing. *Journal of Consulting and Clinical Psychology*, 46(5), 892– 900. https://doi.org/10.1037/0022-006X.46.5.892
- Hoffmann, A. (2019). Mit Betrug in der Pflege verdient man wie im Drogenhandel [Fraud in disability care is financially as lucrative as drug dealing]. Retrieved November 17, 2020, from https://www.stern.de/ wirtschaft/news/mit-betrug-in-der-pflege-verdient-manwie-im-drogenhandel-8570136.html
- Hoffmann-Richter, U., Jeger, J., & Schmidt, H. (2012). Das Handwerk ärztlicher Begutachtung: Theorie, Methodik und Praxis [The craft of independent medical examinations: Theory, methodology, and practice]. Kohlhammer.
- Horner, M. D., VanKirk, K. K., Dismuke, C. E., Turner, T. H., & Muzzy, W. (2014). Inadequate effort on neuropsychological evaluation is associated with increased healthcare utilization. *The Clinical Neuropsychologist*, 28(5), 703–713. https://doi.org/10 .1080/13854046.2014.925143
- Hurtubise, J., Scavoni, A., Sagar, S., & Erdodi, L. A. (2017). Psychometric markers of genuine and feigned neurodevelopmental disorders in the context of applying for academic accommodations. *Psychological Injury and Law*, 10(2), 121–137. https://doi.org/10.1007/s12207-017-9287-5

- Iverson, G. L., Terry, D. P., Karr, J. E., Panenka, W. J., & Silverberg, N. D. (2018). Perceived injustice and its correlates after mild traumatic brain injury. *Journal of Neurotrauma*, 35(10), 1156–1166. https:// doi.org/10.1089/neu.2017.5402
- Jelicic, M., Merckelbach, H., & Bošković, I. (2017). Seven myths about feigning. In H. Otgaar & M. L. Howe (Eds.), *Finding the truth in the courtroom: Dealing with deception, lies, and memories* (pp. 227–242). Oxford University Press. https://doi.org/ 10.1093/oso/9780190612016.003.0011
- Kleideiter, W. (2017, April 25). DRV zeigt Arzt aus Südwestfalen an: Renten-Betrug: 600 neue Verdachtsfälle [German Pension Fund denounced physician from Southwestphalia: Pension fraud in 600 suspected new cases]. Westfälische Nachrichten. Retrieved November 01, 2017, from http://www .wn.de/Muensterland/2781570-DRV-zeigt-Arzt-aus-Suedwestfalen-an-Renten-Betrug-600-neue-Verda chtsfaelle
- Klieve, H., Beamish, W., Bryer, F., Rebollo, R., Perrett, H., & van den Muyzenberg, J. (2010). Assessing practitioner expertise through online survey tool Lime-Survey. Retrieved October 3, 2022, from https://resea rch-repository.griffith.edu.au
- Klumbies, G. (1980). Psychotherapie in der Inneren und Allgemeinmedizin [Psychotherapy in internal and general medicine] (3rd ed.). S. Hirzel.
- Linden, M., Bülau, N. I., Kessemeier, F., Kobelt, A., & Bassler, M. (2020). Humiliation and injustice: Most frequent and most hurting stressors in psychosomatic patients. *Psychological Disorders and Research*, 3(2), 2–6. https://doi.org/10.31487/j.PDR.2020.02.02
- Linden, M., & Rotter, M. (2020). Spectrum of embitterment manifestations. *Psychological Trauma: Theory, Research, Practice, and Policy*, 10(1), 1–6. https://doi.org/10.1037/tra0000307
- Marshall, P. S., Hoelzle, J. B., Heyerdahl, D., & Nelson, N. W. (2016). The impact of failing to identify suspect effort in patients undergoing adult attention-deficit/ hyperactivity disorder (ADHD) assessment. *Psychological Assessment*, 28(10), 1290–1302. https:// doi.org/10.1037/pas0000247
- Martin, P. K., & Schroeder, R. W. (2020). Base rates of invalid test performance across clinical nonforensic contexts and settings. *Archives of Clinical Neuropsychology*, 35(6), 717–725. https://doi.org/ 10.1093/arclin/acaa017
- Martin, P. K., Schroeder, R. W., & Odland, A. P. (2015). Neuropsychologists' validity testing beliefs and practices: A survey on North American professionals. *The Clinical Neuropsychologist*, 29(6), 741–776. https:// doi.org/10.1080/13854046.2015.1087597
- Merckelbach, H., & Merten, T. (2012). A note on cognitive dissonance and malingering. *The Clinical Neuropsychologist*, 26(7), 1217–1229. https://doi.org/ 10.1080/13854046.2012.710252

- Merten, T., & Giger, P. (2018). Wie häufig treten Simulation und Aggravation in der Begutachtung auf? Schätzungen von Laien [Lay persons' prevalence estimates of malingering in independent medical and psychological examinations]. *Forensische Psychiatrie, Psychologie, Kriminologie, 12*(2), 164–171. https://doi.org/10.1007/s11757-018-0465-4
- Merten, T., Kaminski, A., & Pfeiffer, W. (2020). Prevalence of overreporting on symptom validity tests in a large sample of psychosomatic rehabilitation inpatients. *The Clinical Neuropsychologist*, 34(5), 1004–1024. https://doi.org/10.1080/13854046.2019.1694073
- Merten, T., & Merckelbach, H. (2020). Factitious disorder and malingering. In J. R. Geddes, N. C. Andreasen, & G. M. Goodwin (Eds.), *The new Oxford textbook of psychiatry* (3rd ed., pp. 1342–1349). Oxford University Press. https://doi.org/10.1093/me d/9780198713005.003.0132
- Merten, T., & Dandachi-FitzGerald, B. (2022). Symptom and performance validity assessment: European trends in research and practice [Editorial]. *Psychological Injury and Law*, 15(2), 113–115. https://doi.org/10.1007/s12207-022-09454-0
- Mittenberg, W., Patton, C., Canyock, E. M., & Condit, D. C. (2002). Base rates of malingering and symptom exaggeration. *Journal of Clinical and Experimental Neuropsychology*, 24(8), 1094–1102. https://doi.org/10.1076/jcen.24.8.1094.8379
- Nies, K. J., & Sweet, J. J. (1994). Neuropsychological assessment and malingering: A critical review of past and present strategies. *Archives of Clinical Neuropsychology*, 9(6), 501–552. https://doi.org/10.1093/ arclin/9.6.501
- Palena, N., Caso, L., Cavagnis, L., Greco, A., & Vrij, A. (2022). Exploring the relationship between personality, morality and lying: A study based on the person-centred approach. *Current Psychology*. Advance online publication. https://doi.org/10.1007/s12144-022-03132-9
- Puente-López, E., Pina, D., López-López, R., González Ordi, H., Bošković, I., & Merten, T. (2022). Prevalence estimates of symptom feigning and malingering in Spain. *Psychological Injury and Law*. Advance online publication. https://doi.org/ 10.1007/s12207-022-09458-w
- Quezada-Ortega, M. R., Razo-Mondragón, J. L. P., Marín-Cotoñieto, I. A., Salinas-Tovar, S., & López-Rojas, P. (2006). Simulación en trabajadores que solicitan pensión por invalidez laboral [Malingering in workers applying for occupational disability pensions]. *Gaceta Medica de Mexico*, 142(2), 109–112.
- Ramesh, V. (2013). Functional neurological disorders in children and young people. *Developmental Medicine and Child Neurology*, 55(1), 3–4. https:// doi.org/10.1111/dmcn.12058
- Resnick, P. J., West, S., & Payne, J. W. (2008). Malingering of posttraumatic stress disorder. In R. Rogers (Ed.), *Clinical assessment of malingering and deception* (3rd ed., pp. 109–127). Guilford Press.

- Reuber, M., Mitchell, A. J., Howlett, S. J., Crimlisk, H. L., & Grünewald, R. A. (2005). Functional symptoms in neurology: Questions and answers. *Journal of Neurology, Neurosurgery, and Psychiatry*, 76(3), 307–314. https://doi.org/10.1136/jnnp.2004.048280
- Rickards, T. A., Cranston, C. C., Touradji, P., & Bechtold, K. T. (2018). Embedded performance validity testing in neuropsychological assessment: Potential clinical tools. *Applied Neuropsychology: Adult*, 25(3), 219–230. https://doi.org/10.1080/2327 9095.2017.1278602
- Rogers, R. (1990). Models of feigned mental illness. Professional Psychology, 21(3), 182–188. https:// doi.org/10.1037/0735-7028.21.3.182
- Rogers, R., Salekin, R. T., Sewell, K. W., Goldstein, A., & Leonard, K. (1998). A comparison of forensic and nonforensic malingerers: A prototypical analysis of explanatory models. *Law and Human Behavior*, 22(4), 353–367. https://doi.org/10.1023/A:1025 714808591
- Rosenhan, D. L. (1973). On being sane in insane places. Science, 179(4070), 250–258. https://doi.org/10.1126/ science.179.4070.250
- Ruff, R. M., Klopfer, J., & Blank, J. (2016). Homogenous base rates for malingering in neuropsychological examinations of litigants. *Psychological Injury and Law*, 9(2), 143–153. https://doi.org/10.1007/ s12207-016-9259-1
- Santamaría, P., Capilla Ramírez, P., & González Ordi, H. (2013). Prevalencia de simulación en incapacided temporal: Perceptión de los profesionales de la salud [Health professionals' perceptions of prevalence of malingering of temporary disability in Spain]. *Cliníca y Salud*, 24(3), 139–151. https:// doi.org/10.1016/S1130-5274(13)70015-X
- Schlicht, D., & Merten, T. (2014). Das Bild vorgetäuschter Gesundheitsstörungen in der öffentlichen Meinung [The picture of malingered symptom presentations in public opinion]. Versicherungsmedizin, 66(3), 146–152.
- Schroeder, R. W., Clark, H. A., & Martin, P. K. (2022). Base rates of invalidity when patients undergoing routine clinical evaluations have social security disability as an external incentive. *The Clinical Neuropsychologist*, *36*(7), 1902–1914. https://doi.org/10 .1080/13854046.2021.1895322
- Schroeder, R. W., & Martin, P. K. (Eds.). (2022). Validity assessment in clinical neuropsychological practice: Evaluating and managing noncredible performance. Guilford.
- Schmand, B., Lindeboom, J., Schagen, S., Heijt, R., Koene, T., & Hamburger, H. L. (1998). Cognitive complaints in patients after whiplash injury: The impact of malingering. *Journal of Neurology, Neurosurgery, and Psychiatry*, 64(3), 339–343. https:// doi.org/10.1136/jnnp.64.3.339
- Schmand, B., & Ponds, R. (1997). Malingeren: Simuleren en aggraveren [Malingering: Simulation and

symptom exaggeration] In B. Deelman, P. Eling, E. de Haan, A. Jennekens-Schinkel, & E. van Zomeren (Eds.), *Klinische neuropsychologie* (pp. 426–436). Boom.

- Schmidt, T., Lanquillon, S., & Ullmann, U. (2011). Kontroverse zu Beschwerdenvalidierungsverfahren bei der Begutachtung psychischer Störungen [Controversy of symptom validity assessment in forensic evaluation of mental disorders]. Forensische Psychiatrie, Psychologie, Kriminologie, 5(3), 177–183. https://doi.org/10.1007/s11757-011-0113-8
- Schneider, B. C., Schöttle, D., Hottenrott, B., Gallinat, J., & Moritz, S. (2019). Assessment of adult ADHD in clinical practice: Four letters—40 opinions. *Journal of Attention Disorders*. Advance online publication. https://doi.org/10.1177/1087054719879498
- Streppel, M., & Brusis, T. (2010). Zur Problematik der Simulation und Aggravation in der HNO-ärztlichen Begutachtung [The problem of malingering and symptom exaggeration in independent otorhinolaryngological examinations]. *HNO*, 58(2), 126–131. https://doi.org/10.1007/s00106-009-2064-3
- Suhr, J. A., & Johnson, E. E. (2022). First do no harm: Ethical issues in pathologizing normal variations in behavior and functioning. *Psychological Injury and Law*, 15(3), 253–267. https://doi.org/10.1007/s12207-022-09455-z
- Sweet, J. J. (2009). Neuropsychology and the law: Malingering assessment in perspective. In J. E. Morgan & J. J. Sweet (Eds.), *Neuropsychology of malingering casebook* (pp. 3–8). Psychology Press.
- Tucha, L., Fuermaier, A. B. M., Koerts, J., Groen, Y., & Thome, J. (2015). Detection of feigned attention deficit hyperactivity disorder. *Journal of Neural Transmission*, 122(Suppl. 1), S123–S134. https:// doi.org/10.1007/s00702-014-1274-3
- van der Heide, D., Bošković, I., van Harten, P., & Merckelbach, H. (2020). Overlooking feigning behavior may result in potential harmful treatment

interventions: Two cases of undetected malingering. Journal of Forensic Sciences, 65(4), 1371–1375. https://doi.org/10.1111/1556-4029.14320

- van Helvoort, D., Merckelbach, H., van Nieuwenhuizen, C., & Otgaar, H. (2022). Traits and distorted symptom presentations: A scoping review. *Psychological Injury and Law*, 15(2), 151–171. https:// doi.org/10.1007/s12207-022-09446-0
- Ward, M. K., & Meade, A. W. (2022). Dealing with careless responding in survey data: Prevention, identification, and recommended best practices. *Annual Review of Psychology*, 74(1), 577–596. https://doi.org/10.1146/annurev-psych-040422-045007
- Weiss, R., & Rosenfeld, B. (2010). Cross-cultural validity in malingering assessment: The Dot Counting Test in a rural Indian sample. *International Journal of Forensic Mental Health*, 9(4), 300–307. https://doi.org/10.1080/14999013.2010 .526680
- Widder, B., Berchtold, J., Foerster, K., Gaidzik, P. W., Henningsen, P., & Schneider, W. (2016). Standards der Begutachtung bei der Beurteilung von Kausalitätsfragen [Standards of independent medical examinations with respect to causality]. In W. Schneider, R. Doh - renbusch, HJ. Freyberger, P. Henningsen, H. Irle, V. Köllner, & B. Widder (Eds.), *Begutachtung bei psychischen und psychosomatischen Erkrankungen. Autorisierte Leitlinien und Kommentare* (2nd ed., pp. 549–621). Hogrefe.
- Young, G. (2015). Malingering in forensic disabilityrelated assessments: Prevalence 15 ± 15 %. Psychological Injury and Law, 8(3), 188–199. https:// doi.org/10.1007/s12207-015-9232-4

Received October 15, 2022 Revision received November 29, 2022 Accepted January 9, 2023